Author's response to reviews

Title: Extent and Distribution of the Air Column and Symptoms of Gastroesophageal Reflux Disease

Authors:

Alijavad Mousavi M.D (MousaviAlijavad@yahoo.com)
Hanie Raji M.D (RajiHanie@yahoo.com)
Mojtaba Teimoori M.D (mojtaba_teimoori@irimc.ir)
Shadi Ghourchian M.D (shadighourchian@yahoo.com)

Version: 2 Date: 25 May 2011

Author's response to reviews: see over
Extent and Distribution of the Air Column and Symptoms of Gastroesophageal Reflux Disease

Mousavi Alijavad MD ¹ (MousaviAlijavad@yahoo.com)
Raji Hanie MD ² (RajiHanie@yahoo.com)
Teimoori Mojtaba MD ³ (mojtaba_teimoori@irimc.org)
Ghourchian Shadi ⁴ (shadighorchian@yahoo.com)

¹- Associate Professor of lung diseases, Rasoul-e-Akram Hospital, Tehran University of Medical Sciences, Tehran, Iran
²- Associate Professor of lung diseases, Jundishapur University of Medical Sciences, Ahvaz, Iran.
³- Nephrology and urology research Center, Medical Faculty, Bagyataallah University of Medical Sciences, Tehran, Iran.
⁴- Medical student, Medical Student Research Committee of Tehran University of Medical Sciences, Tehran, Iran (Corresponding Author)

Correspondence: Ghourchian Shadi, Medical student, Medical Student Research Committee of Tehran University of Medical Sciences, Tehran, Iran E-mail (shadighorchian@yahoo.com)

This study was done in Tehran – Rassoule Akram Hospital – Tehran University of Medical Science.

Dear editors BMC

Working on novel ideas and publishing in leading specialized journals have been of our perspectives in designing and performing the submitted study.

The authors did their best to consider almost all aspects of technical, practical and ethical issues in this trial. We disclose all financial and personnel relationship that might bias my work.

Authors state explicitly there are no conflicts about this paper. Rights of patients were respected considering Helsinki declaration. The study is not published elsewhere.

Authors of this study had full excess to all the data in the study and we take complete responsibility for the integrity of the data and the accuracy of the data analysis.

Your journal as one of the pioneers in publishing articles in the area of Radiology persuades us to ask your warm cooperation for entering our work in the process of review and publication. The attempts and expertise of the authors in one of the top universities of our country seems to be a good reason for this request.

We hope to take a positive step in science.

Sincerely,

Authors of the article